Author's response to reviews

Title: An evaluation of the effectiveness of a community mentoring service for socially isolated older people: a case-matched controlled trial

Authors:

Andy P Dickens M.Sc. (andy.dickens@pms.ac.uk)
Suzanne H Richards Ph.D. (suzanne.richards@pms.ac.uk)
Annie Hawton M.Sc. (annie.hawton@pms.ac.uk)
Rod S Taylor Ph.D. (rod.taylor@pms.ac.uk)
Colin J Greaves Ph.D. (colin.greaves@pms.ac.uk)
Colin Green Ph.D. (colin.green@pms.ac.uk)
Rachel Edwards M.Sc. (rachel.edwards1@awp.nhs.uk)
John L Campbell M.D. (john.campbell@pms.ac.uk)

Version: 2 Date: 9 September 2010

Author's response to reviews: see over
Dear Dr Norton

I am submitting a revised set of documents for our submission, correcting the presentation of tables and figures.

I originally uploaded Tables 1, 2, and 3 as ‘figures’ rather than as ‘additional files’, which has now been corrected. I have also attached captions to each individual figure and table file as well as listing them in the manuscript, and have moved the table legends from the manuscript into the relevant files.

Please accept my apologies for the revision, and I look forward to hearing from you.

Yours sincerely

Andy Dickens